Recommended summary plan for emergency care and treatment: ReSPECT a mixed-methods study

Gavin D Perkins,^{1,2*} Claire A Hawkes,¹ Karin Eli,³ James Griffin,¹ Claire Jacques,¹ Caroline J Huxley,³ Keith Couper,^{1,2} Cynthia Ochieng,⁴ Jonathan Fuld,⁵ Zoe Fritz,⁵ Rob George,⁶ Doug Gould,⁷ Richard Lilford,³ Martin Underwood,¹ Catherine Baldock,⁸ Chris Bassford,⁸ Peter-Marc Fortune,⁹ John Speakman,² Anna Wilkinson,¹⁰ Bob Ewings,^{1,11} Jane Warwick,¹ Frances Griffiths³ and Anne-Marie Slowther³

Declared competing interests of authors: Gavin D Perkins reports grants from the National Institute for Health and Care Research (NIHR), during the conduct of the study; was an executive committee member of Resuscitation Council UK (London, UK) (2009 to present) and Recommended Summary Plans for Emergency Care and Treatment (ReSPECT) committee member of the Resuscitation Council UK (2015 to present); was a board member of the NIHR Programme Grants for Applied Research (2016-20), Health Technology Assessment (HTA) (2020 to present) and CTU Standing Advisory Committee (2019 to present); and is a NIHR senior investigator. Karin Eli reports grants from NIHR, during the conduct of the study. Keith Couper reports grants from NIHR, during the conduct of the study, and grants from the Resuscitation Council UK, outside the submitted work. Zoe Fritz chairs the ReSPECT subcommittee for the Resuscitation Council UK and sits on the executive committee for the Resuscitation Council UK. Martin Underwood reports grants from NIHR during the conduct of the study; personal fees from NIHR for acting as journal editor and member of the Journal Editors Group, grants from NIHR for being chief investigator and co-investigator on multiple research grants, grants from Versus Arthritis (Chesterfield, UK); other from multiple professional conferences, grants from Stryker Ltd (Newbury, UK), grants from Serco Group plc (Hook, UK), other from Confederation for Advanced Research Training in Africa (Nairobi, Kenya) and personal fees from Clinvivo Ltd (Kent, UK),

¹Warwick Clinical Trials Unit, University of Warwick, Coventry, UK

²University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

³Warwick Medical School, University of Warwick, Coventry, UK

⁴University of Bristol, Bristol, UK

⁵Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK

⁶St Christopher's Hospice, London, UK

⁷Intensive Care National Audit & Research Centre, London, UK

⁸University Hospitals of Coventry and Warwickshire NHS Trust, Coventry, UK

⁹Manchester University NHS Foundation Trust, Manchester, UK

¹⁰Hampshire Hospitals NHS Foundation Trust, Hampshire, UK

¹¹Patient and public contributor

^{*}Corresponding author G.D.Perkins@warwick.ac.uk

outside the submitted work; is an associate investigator on multiple research grants from the Australian National Health and Medical Research Council (Canberra, ACT, Australia); and was part of the NIHR Journals Library Editors Group (2016–19) and of the HTA Commissioning Committee (2010–14). Catherine Baldock was the clinical lead for ReSPECT implementation at the Resuscitation Council UK. Bob Ewings received funding from the study for his PPI work and was also a member of the ReSPECT National Working Group. Jane Warwick reported grants from NIHR, during the conduct of the study. Frances Griffiths reported grants from NIHR, during the conduct of the study. Anne-Marie Slowther reports grants from the NIHR Health and Social Care Delivery Research programme (references 13/10/14 and 17//99/34) and the HTA programme (reference 16/77/02), outside the submitted work; and was co-author on an evidence synthesis of the do not attempt cardiopulmonary resuscitation decision-making that informed the development of the ReSPECT process.

Published December 2022 DOI: 10.3310/LFPE3627

Plain English summary

ReSPECT mixed-methods study

Health and Social Care Delivery Research 2022; Vol. 10: No. 40

DOI: 10.3310/LFPE3627

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

o not attempt cardiopulmonary resuscitation decisions have been criticised for not involving patients and for being unclear about what other care and treatment is required. The Recommended Summary Plan for Emergency Care and Treatment (ReSPECT) process aims to help health-care professionals, patients and families to make better decisions, together, about emergency treatments, including cardiopulmonary resuscitation.

This study examined how, when and why ReSPECT plans are made in hospital and how they affect patient care. We talked to hospital doctors, nurses, patients, families and general practitioners. In addition, we observed ReSPECT conversations taking place in hospitals, examined patients' medical records and looked at how the ReSPECT process affected what happened to patients.

The study took place between 2017 and 2020, soon after the ReSPECT process started being used in hospitals. By December 2019, the ReSPECT process was being used in nearly one in four hospitals. Only one in five patients in hospital, usually those who were most unwell, had a ReSPECT recommendation. Doctors told us that they lacked time to talk to more patients about the ReSPECT process.

Most conversations we observed were about resuscitation, but conversations also covered other treatments. Some conversations included discussion about what mattered to patients. When a doctor was clear about how well a treatment would work or not, the doctor would usually try to persuade the patient to agree with their recommendations. If a doctor was less certain about a treatment, then conversations were more open and explored the patient's wishes to a greater extent.

Hospital doctors and general practitioners gave different views on the purpose of the ReSPECT process and the type of recommendations they would record.

Further research is needed to understand how the NHS can engage more people in conversations about their future emergency care and treatment. Research should explore how doctors, patients and families can best work together to reach shared decisions.

Health and Social Care Delivery Research

ISSN 2755-0060 (Print)

ISSN 2755-0079 (Online)

Health and Social Care Delivery Research (HSDR) was launched in 2013 and is indexed by Europe PMC, DOAJ, INAHTA, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

This journal was previously published as *Health Services and Delivery Research* (Volumes 1–9); ISSN 2050-4349 (print), ISSN 2050-4357 (online)

The full HSDR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr.

Criteria for inclusion in the Health and Social Care Delivery Research journal

Reports are published in *Health and Social Care Delivery Research* (HSDR) if (1) they have resulted from work for the HSDR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HSDR programme

The HSDR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HSDR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-and-social-care-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HSDR programme or one of its preceding programmes as project number 15/15/09. The contractual start date was in November 2016. The final report began editorial review in September 2021 and was accepted for publication in June 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HSDR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care.

Copyright © 2022 Perkins *et al.* This work was produced by Perkins *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk